Author's response to reviews

Title: Caregiver burden in amyotrophic lateral sclerosis is more dependent on patients' behavioral changes than physical disability: a comparative study.

Authors:

   Patricia Lillo (pclz@uchile.cl)
   Eneida Mioshi (e.mioshi@neura.edu.au)
   John R Hodges (J.hodges@neura.edu.au)

Version: 2 Date: 3 April 2012

Author's response to reviews: see over
Dear Dr. Majithia,

Re: Caregiver burden in amyotrophic lateral sclerosis is more dependent on patients’ behavioral changes than physical disability: a comparative study.

Authors: Patricia Lillo, Eneida Mioshi, John R. Hodges.

Please find attached a copy of the above manuscript, which I ask your kind consideration as an article for publication in BMC Neurology. We think our study is suitable for publication because it describes the physical and behavioral features of a large cohort of patients with amyotrophic lateral sclerosis as well as their impact on caregiver burden. Importantly, it also takes into account the caregivers’ factors implied on caregiver burden.

No material submitted as part of a manuscript infringes existing copyrights, or the rights of a third party.

Yours sincerely,

Dr Patricia Lillo
Neuroscience Research Australia
Barker Street, Randwick, NSW 2031 AUSTRALIA
PO Box 82, St Pauls, NSW 2031 AUSTRALIA